Author's response to reviews

Title: Soy foods have low glycemic and insulin response indices in normal weight subjects

Authors:

Robert M Blair Ph.D. (drblair@revivalsoy.com)
E. C. Henley Ph.D. (henleyec@aol.com)
Aaron Tabor M.D. (drtabor@revivalsoy.com)

Version: 3 Date: 15 December 2006

Author's response to reviews:

The second set of revisions have been made in the abstract as requested. The statistical analysis method has been removed and the number of test foods in each experiment has been added.